Dear editor;

Thank you for your interest; we have read and considered your advices and cautions. We have made the changes you requested as follows:

1. We changed the title as you advised (A presenting with B in C: a case report).
2. We placed the information you requested in the title page.
3. We have given the ethnicity of the patient both in the abstract and manuscript.
4. Written informed consent was obtained from the patient for publication of this case report and accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.
5. The authors declare that they have no competing interests.
6. We reported the contributions of each author to the manuscript.
7. All authors have read and approved the final manuscript.

Thank you very much for your consideration of this manuscript.

Sincerely yours.

AUTHORS:
Gonul Çiçek-Sentürk (drgonulcicek@hotmail.com)
Fatma Aybala Altay (aybalaaltay@hotmail.com)
Ayşegül Ulu-Kılıç (draysegululu@yahoo.com)
Yunus Gurbuz (gurbuzyunus@hotmail.com)
Ediz Tutuncu (ediztutuncu@gmail.com)
İrfan Sencan (isencanibuy@yahoo.com)